Author's response to reviews

Title: Title: Problem-based learning in resource-poor settings: lessons from a medical school in Ghana

Authors:

Daniel Amoako-Sakyi (d.amoako-sakyi@uccsms.edu.gh)
Harold S Amonoo-Kuofi (hsak@uccsms.edu.gh)

Version: 2
Date: 27 January 2015

Author's response to reviews:

Dear Editor,

Below is an outline of revisions made to manuscript.

1. An Acknowledgements section has be included in the manuscript.
2. The last sentence of the background section of the abstract has be modified.

Thank you.

*Daniel Amoako-Sakyi